Contents lists available at ScienceDirect





journal homepage: www.elsevier.com/locate/jie

Factor Intensity, product switching, and productivity: Evidence from Chinese exporters $\overset{\curvearrowleft}{\sim}$



Yue Ma^a, Heiwai Tang^b, Yifan Zhang^{c,*}

^a Department of Economics and Finance, City University of Hong Kong, 83 Tat Chee Avenue, Kowloon Tong, Kowloon, Hong Kong

^b School of Advanced International Studies, Johns Hopkins University, 1717 Massachusetts Ave NW, Washington, DC 20036, USA

^c Department of Economics, Lingnan University, 8 Castle Peak Road, Tuen Mun, Hong Kong

ARTICLE INFO

Article history: Received 21 January 2008 Received in revised form 30 August 2013 Accepted 5 November 2013 Available online 14 November 2013

JEL classification: F11 L16 Q53

Keywords: Exporters Productivity Factor intensity Multi-product firms

1. Introduction

A rich body of research documents the superior performance of exporters compared to non-exporters. Exporters are larger, more productive, more capital-intensive, and more skill-intensive (e.g., Bernard and Jensen, 1999). Existing explanations for exporters' superior performance can be categorized into three broad themes: self-selection (e.g., Clerides et al., 1998; Bernard et al., 2003; Melitz, 2003), learning by exporting (e.g., Aw et al., 2000; Van Biesebroeck, 2005; De Loecker, 2007), and firms' investment in preparation for export (e.g., Bernard and Jensen, 1997; Yeaple, 2005; Lileeva and Trefler, 2010; Bustos, 2011; Aw et al., 2011; Iacovone and Javorcik, 2012).

E-mail addresses: yue.ma@cityu.edu.hk (Y. Ma), hwtang@jhu.edu (H. Tang), yifan.zhang@ln.edu.hk (Y. Zhang).

ABSTRACT

This paper analyzes how a firm's specialization in its core products after exporting affects its factor intensity and productivity. Using Chinese manufacturing firm data for the 1998–2007 period, we find that firms become less capital-intensive but more productive after exporting, compared to non-exporters that share similar ex ante characteristics. To rationalize these findings that contrast with existing studies, we develop a variant of the model by Bernard, Redding, and Schott (2010, 2011) to consider firms producing multiple products with varying capital intensity. The model predicts that when a firm in a labor-abundant country starts exporting, it specializes in its core competencies by allocating more resources to produce more labor-intensive products. Firm ex ante productivity is associated with a smaller decline in capital intensity after exporting. A sharper post-export decline in capital intensity is associated with a larger increase in measured total factor productivity. We find firm-level evidence supporting these predictions. Using transaction-level data for the 2000–2006 period, we show that Chinese new exporters add products that are less capital-intensive than their existing products and drop those that are more capital-intensive in subsequent years.

© 2013 Elsevier B.V. All rights reserved.

This paper studies a lesser known effect of exporting on firm performance: how a firm's specialization in its core products after exporting affects its factor intensity and thus productivity. Using a large panel data set of China's manufacturing firms for the 1998-2007 period, we find that firms' measured productivity increases but capital intensity declines after exporting. Figs. 1 (unbalanced panel) and 2 (balanced panel) show that, although the average firm capital intensity increased for both exporters and non-exporters from 1998 to 2007 in China, exporters were persistently less capital-intensive than non-exporters and there was no sign of convergence before 2007. We confirm that exporters have a relatively lower capital intensity than non-exporters, both within firms and within a narrowly defined industry, and for both domestic and foreign firms. To tackle the potential estimation bias due to firms' selection into exporting, we use various matching methods to compare exporters and non-exporters with similar ex ante characteristics (i.e., Heckman et al., 1997 and subsequent studies).¹ Within the same bins of ex ante productivity, capital intensity and sales, we find that new exporters experienced a significant decline in capital intensity relative to non-exporters. Moreover, we show that the relative decline in capital intensity after exporting is smaller for the ex ante more productive exporters, but is larger for the ex ante more capital-intensive ones.

[★] We thank the editor, Daniel Trefler, and two anonymous referees for their valuable comments. We also thank Daniel Berkowitz, Loren Brandt, Swati Dhingra, Chang-Tai Hsieh, Yasheng Huang, Amit Khandelwal, Steven Klepper, Alexis Leon, Ben Mandel, Marc Muendler, Soiliou Namoro, Emanuel Ornelas, Albert Park, Veronica Rappaport, Thomas Rawski, Stephen Redding, Peter Schott, Robert Staiger, Jo Van Biesebroeck, and seminar participants at the AEA Meetings, Berkeley-Booth-IGC China Summer Institute, Clemson, Hitotsubashi-COE Conference, IFPRI, Johns Hopkins SAIS, LSE, LMU Munich, NBER ITI Meetings, Penn-State-Tsinghua Conference, World Bank, and UIUC, for their helpful comments and discussions. The authors gratefully acknowledge funding from Competitive Earmarked Research Grant (LU3409/06H) of the RGC of HKSAR Government. The usual disclaimer applies.

^{*} Corresponding author. Tel.: +852 2616 7149.

^{0022-1996/\$ -} see front matter © 2013 Elsevier B.V. All rights reserved. http://dx.doi.org/10.1016/j.jinteco.2013.11.003

¹ In this paper, we address endogeneity issues by using the matching estimation techniques only. We do not have well-defined instrumental variables, such as tariff cuts as in Lileeva and Trefler (2010).



Fig. 1. Note: the unbalanced panel includes all firms in our sample. Source: China's NBS above-scale manufacturing firm data.



Fig. 2. Note: the balanced panel includes only those firms that appear every year in our sample. The balanced sample covers 7.6% of all firms in our sample. Source: China's NBS above-scale manufacturing firm data.

Our findings regarding the relatively lower capital intensity among exporters contrast sharply with the existing evidence from both developed and developing countries (e.g., Bernard and Wagner, 1997; Bernard and Jensen, 2004; Van Biesebroeck, 2005; De Loecker, 2007). However, our results provide "mirror image" evidence supporting Bernard et al. (2006), who find that U.S. manufacturing firms in sectors facing more import competition from low-wage countries are more likely to switch industries and become more skill and capital-intensive over time. During our sample period, we find that Chinese firms exploited the country's comparative advantage and used less capital in production when they became exporters. These findings show that the classic Heckscher–Ohlin forces are operating at the firm level, serving as another channel through which trade can affect the factor markets in both developing countries and their trade partners.²

What accounts for the decline in firms' capital intensity after exporting and the heterogeneous outcomes across firms? To answer this question, we develop a variant of the multi-product model by Bernard et al. (2010, 2011) to consider both capital and labor as factors of production. In the model, heterogeneous firms can produce a continuum of products with different capital intensities of production. Besides firm heterogeneity in productivity ("ability") as in Melitz (2003), a firm's profitability from selling a product in a foreign market depends on a set of exogenous firm-product "consumer taste" attributes. In addition to the country-specific fixed export cost, exporting an additional product entails extra fixed costs (e.g., R&D expenditure to produce a blue print or overhead cost to manage a product-specific sales team). Thus, a firm will export a product only if its product attributes guarantee sufficiently high revenue to cover these extra fixed costs. Given China's labor abundance, labor-intensive products are on average associated with lower zero-profit thresholds than capitalintensive products for all firms. When a firm receives a favorable cost shock and starts exporting to a capital-abundant country, it will specialize in its core competencies - its labor-intensive products. Thus, a firm becomes more labor-intensive after exporting either by expanding sales of existing labor-intensive products (the intensive margin) or by adding more labor-intensive products (the extensive margin).

Although our model focuses on product churning as a driver of the observed change in firm factor intensity, this is not the only channel through which exporting has an effect. Firms may still invest in new capital or capital-intensive activities, as shown by some existing studies (e.g., Aw et al., 2011; Bustos, 2011). It should be noted that our empirical results show the net effect of exporting on capital intensity, implying that in China the standard investment effects proposed in the literature

² Although Bernard et al. (2007b) also embed a Heckscher–Ohlin framework within the Melitz (2003) model, firms in their model only differ in terms of productivity but not factor intensity within an industry.

could be dominated by product-churning effects. Importantly, our model is general enough to be extended to incorporate the investment decisions considered by existing studies for an examination of the gross effects of product churning through different channels.

Our model yields two predictions in addition to rationalizing the findings on exporters' lower capital intensity. Firstly, it shows how changes in product scope can affect a firm's measured total-factor productivity (TFP). In particular, firms that have a larger reallocation of resources from capital-intensive to labor-intensive products after exporting have a bigger increase in measured TFP. Given fixed export costs and firm productivity, an increase in the sales of labor-intensive products implies a larger extent of increasing returns relative to capital-intensive products. Secondly, our model also predicts that the ex ante more productive exporters experience less product churning and have a smaller decline in capital intensity after exporting. We find supporting evidence for both theoretical predictions using firm-level data. Although recent studies on multi-product firms have made similar predictions, to the best of our knowledge there is little direct evidence of the effect of trade through the proposed product churning channel on firm productivity.³ Our findings provide a new angle for interpreting the relationship between exporting and firm productivity, both ex ante and ex post, in addition to the existing empirical studies that focus mainly on learning by exporting or selection.

To provide further evidence of product churning's effect on firm's factor intensity after exporting, we use transaction-level trade data merged with manufacturing firm data. We compute the weighted average capital intensity of a product (HS 6-digit), using the capital intensity of the firms producing that product. Employing these product-level capital intensity measures, we investigate whether firms' product churning patterns are consistent with the model prediction that firms specialize in labor-intensive products after exporting. We find that over the 2000–2006 period, new exporters in China added products that are less capital-intensive than their existing products and dropped those that are more capital-intensive in the year following the first year of exporting. The newly added products were on average more labor-intensive if the destination country was more capital-abundant.

The rest of the paper is organized as follows. Section 2 reviews the related literature. Section 3 describes our data source. Section 4 explores the basic patterns of export participation, capital intensity, and productivity. Section 5 examines the effect of exporting on new exporters' capital intensity. Section 6 presents a theoretical model to rationalize our findings. Sections 7 and 8 examine the specific theoretical predictions using transaction-level trade data. The last section concludes.

2. Related literature

Our paper relates to several strands of the literature. First, it contributes to the growing theoretical literature on multi-product exporters. A common feature of this literature is that diversification across products is costly and that access to foreign markets provides an opportunity for firms to specialize in their core competencies. Feenstra and Ma (2008) study how trade liberalization reduces firms' product scope due to the presence of cannibalization effects. Nocke and Yeaple (2008) study the implications when a firm's marginal cost of production increases in product scope due to managers' limited span of control as in Lucas (1978). Eckel and Neary (2010) develop a model featuring each firm having a core product that is associated with the lowest marginal cost, with increasing marginal costs for products farther away from the core. Bernard et al. (2011) show that trade

liberalization can theoretically result in both within and across-firm reallocation of resources, leading to growth in both firm and aggregate productivity. The added multi-product dimension permits firms to drop products that are less appealing to consumers and add those that are more appealing upon trade liberalization, raising measured firm productivity. The related empirical literature is also growing. Using U.S. publicly listed firms data, Liu (2010) shows that firms are more likely to drop peripheral products to refocus on core competencies in response to trade liberalization. Arkolakis and Muendler (2011) document the within-firm pattern of specialization in core competencies for Brazilian firms. They rationalize their findings using a multiproduct model that features local entry costs that increase in the firms' product scope. Mayer et al. (2012) go further to show the relationship between destination market toughness and exporters' product mix. They show theoretically and empirically that firms increase sales of the best-performing products in tougher markets.

We provide evidence for the positive relationship between product churning and measured firm productivity, by using the change in capital intensity after exporting to capture the degree of specialization. Moreover, we extend the existing multi-product framework, which largely focuses on a single factor of production, to consider both capital and labor as inputs. We empirically verify that specialization in core competencies (labor-intensive products for exporters in developing countries) is associated with a higher measured firm productivity.⁴

Second, as mentioned in the introduction, our paper is related to the existing studies that consistently find exporters to be more capital- or skill-intensive (e.g., Bernard and Wagner, 1997; Bernard and Jensen, 2004; Van Biesebroeck, 2005; De Loecker, 2007). Although most of these studies are silent about the specific channels through which exporting affects firms' factor intensity, the conventional wisdom in the literature is that only the most productive firms select into exporting and that firms invest to upgrade their product quality or production technology before exporting. To the extent that firms' capability is positively correlated with skill or capital intensity (e.g., Yeaple, 2005; Harrigan and Reshef, 2012) or investments are skill and capital-biased (e.g., Verhogeen, 2008), the literature concerning the productivity premium of exporters can readily be used to explain higher capital and skill intensity of exporters. Our findings of lower capital intensity among Chinese exporters should not be taken as a rejection to the investment hypothesis. Instead, our findings imply that product churning appears to dominate the investment and selection mechanisms in China, which leads to a net decline in capital intensity among exporters.

Third, our paper contributes to the extensive literature on the productivity effects of exporting. Previous studies hypothesize that exporters can learn from foreign buyers about product designs and advanced production technology (World Bank, 1993; De Loecker, 2007). Firm-level empirical studies find mixed results for this learning-by-exporting hypothesis (e.g., Clerides et al., 1998; Bernard and Jensen, 1999; Van Biesebroeck, 2005; De Loecker, 2007).⁵ Specific to China, Kraay (1999) finds that exporters are more productive than non-exporters based on survey data of over 2000 firms. Park et al. (2010) use exposure to the 1997 Asian financial crisis as an instrument and find that Chinese firms that export to developed countries experience

³ For example, Bernard et al. (2010) and Mayer et al. (2012) theoretically show that a firm's specialization in core competencies can enhance its measured firm productivity. They provide consistent but not direct evidence for the model predictions.

⁴ In the appendix of Bernard et al. (2010), the authors extend the baseline model to consider two factors of production. They further show how endogenous product choices upon export participation affect firm measured productivity. Our paper extends their model by explicitly solving for how relative factor endowment of the exporting country can serve as a source of within-firm comparative advantage.

⁵ Clerides et al. (1998) and Bernard and Jensen (1999) are among the first studies to empirically distinguish the causal effect of exporting on productivity and self-selection into exporting. They find that exporters have higher productivity than non-exporters before exporting but not after. Other studies find more positive results. A more recent study by a group of economists (International Study Group on Exports and Productivity, 2008) uses comparable firm panel data for 14 countries and an identical method to investigate the relationship between exports and productivity. They find strong evidence for self-selection but no evidence for learning-by-exporting.

higher productivity. Notably, our work is related to Lu (2012), who also incorporates factor intensity in a heterogeneous-firm trade model. While she focuses on export participation across sectors with different capital intensity, we focus on within-firm product churning and its effects on firm performance. She finds that exporters have lower labor productivity than non-exporters due to the lower productivity cutoff for exporting in labor-intensive sectors, whereas we find that exporters have higher measured TFP. The findings that exporters are less capitalintensive in China can reconcile the drastic difference between our results.

Recent studies focus on firms' decisions to simultaneously invest and export. For instance, Lileeva and Trefler (2010) use the elimination of the U.S. tariffs as an instrument to predict Canadian firms' entry into the U.S. market. They show that access to foreign markets enhances labor productivity and technology adoption for less productive firms. Bustos (2011) finds that Argentinian firms that experience sharper tariff cuts in Brazil increase investment in process innovation. Aw et al. (2011) structurally estimate a dynamic model to examine the complementary effects of exporting, investment in technology, and firm productivity. In comparison, Kasahara and Lapham (2013) structurally estimate a dynamic model to examine the complementary effects of exporting, importing, and firm productivity. Instead of focusing on these relationships, we focus on product churning,

3. Data

We use two data sets for the empirical analysis: the above-scale manufacturing firm panel data from China's National Bureau of Statistics (NBS) surveys and customs transactions-level trade data. The manufacturing firm data cover all state-owned firms and all non-stateowned firms with sales above 5 million yuan over the 1998-2007 period (about 0.6 million USD during the sample period).⁶ This data set contains detailed balance-sheet information, such as ownership, output, value added, four-digit industry code (484 categories), exports, employment, original value of fixed asset, and intermediate inputs. The firms in the sample account for 57% of the total industrial value added in 1998 and 94% of that in 2007.7 We exclude observations with missing values for key variables and those that fail to satisfy some basic error checks.⁸ The cleaned data set provides an unbalanced panel of firms that increases in coverage from 148,685 firms in 1998 to 313,048 in 2007.

We use unique numerical IDs to link firms in the sample over time. Firms occasionally receive a new ID as a result of a restructuring, merger, or acquisition. Where possible, we aim to track firms as their boundaries or ownership structures change, linking firms with information such as a firm's name, industry, and address.⁹ These matches are important as one-sixth of all firms that are observed for more than one year experience a change in their official ID over the period of analysis.

In the latter part of the paper, we use transactions-level trade data from China Customs that cover all transactions of Chinese exporters and importers from 2000 to 2006. The trade data provide information on import and export values, quantities, and prices between China and over 200 destination countries at the HS 6-digit level for each trading firm, by ownership of enterprise, and customs regime (ordinary trade and processing trade).¹⁰ As an example, Appendix Table A7 shows the HS 6-digit products within the industry of "footwear, gaiters, and the like" (HS2 = "64"). The purpose of using the transaction-level trade data set is to study within-firm product churning after a firm starts exporting. We merge the manufacturing firm data with the transactionlevel trade data based on firm names and their contact information.¹¹ Table A6 in the Appendix reports statistics of the merged data set. Using the merged data, we can identify new exporters in the trade data set and construct capital intensity measures at the product level (HS6).¹² See Appendix A.3 for details.

A firm's capital intensity is defined as the real value of the capital stock per worker. It is crucial to measure both firm capital and labor accurately. For capital stock, the NBS data only report the original value of fixed asset (OVFS) and net value of fixed asset (NVFS). OVFS is the total capital stock at original purchase prices, while NVFS is OVFS less accumulated depreciation. Thus, OVFS and NVFS are nominal values from different years and cannot be used directly as measures of capital stock. To construct firm capital stock series correctly, we adopt the perpetual inventory method proposed by Brandt et al. (2012). Specifically, we first estimate the firm's initial capital stock using information from its founding year. Then we use the firm's annual investment and assumed depreciation rates to calculate its real capital stock in each year. Appendix A.1 provides the detailed procedure for this approach.¹³ To test the robustness of our results, we also use the NVFS deflated by the industry-specific investment price index as an alternative measure of real capital stock. The investment price indices are taken from various issues of China Statistical Yearbook. From the same source, we also obtain the consumer price index to calculate average real wages at the firm level.¹⁴

As an attempt to adjust for the quality of workers employed by a firm, we use a firm's total wage bill instead of its employment to compute an alternative measure of labor. The problem with this approach is that it is likely to underestimate the total employee compensation, which should also include employee supplementary benefits (Qian and Zhu, 2012).¹⁵ The magnitude of the underestimation may vary across ownership types, regions, and years. Therefore, we use employment as our primary measure for labor and only use total wage bill for a robustness check.

A firm's real output and value added are deflated by a sector-specific ex-factory price index.¹⁶ We use these firm variables to estimate revenue-based TFP. To deal with the estimation biases arising from endogenous input choices, we adopt the Levinsohn and Petrin (2003) procedure that uses intermediate inputs as a proxy for unobservable productivity shocks.¹⁷ For reasons that will become clear below, exporters and non-exporters can have a different factor intensity of production within a disaggregated sector. We thus assume different sector-specific production functions for exporters and non-exporters when estimating

⁶ The unit of analysis is a firm, not the individual plant, but other information in the survey suggests that more than 95% of all observations in our sample are single-plant firms. We focus on manufacturing and exclude mining and utility industries.

⁸ Some firms have missing observations for variables required to calculate productivity. This arises either because the information was not originally reported, or because of negative values for variables such as the real capital stock or value added. Following Jefferson, Rawski and Zhang (2008), we drop all firms with less than eight employees as they fall under a different legal regime. As a result, 17% of firms in the original data set are dropped from the sample in 1998. The percentage excluded drops to 6% in each year after 2001.

The fraction of firms in a year that can be linked to a firm in the previous year increases over time from 84.5% in the first two years (1998-1999) to 92.2% in the final two years (2006-2007). Overall, 95.9% of all year-to-year matches are constructed using firm IDs, and 4.1% using other information on the firm.

¹⁰ The data also include information on quantity, units of quantity, customs offices (ports) where the transactions were processed, and transportation modes.

As shown in Section 8, depending on the year, 37 to 49% of the export value in the trade data set is successfully merged to the NBS firm data set.

¹² To the best of our knowledge, Bernard et al. (2010) are the only group of researchers who did the similar data construction before. They compute the measures of factor intensity at the SIC 5-digit level for the US, and find substantial within-sector (2-digit) heterogeneity in capital and skill intensity. ¹³ The correlation between the two capital stock measures is as high as 0.95.

¹⁴ The price indices are from China Statistical Yearbook, various issues.

¹⁵ In our data, labor's share of value added is only 34%, which is much lower than the 55 to 60% suggested by national income accounting, Hsieh and Klenow (2009) also experience the same problem with the same data set. They assume that non-wage benefits are a constant fraction of a plant's wage bill and select an adjustment factor so that the wage plus non-wage compensation equals half of China's aggregate value added.

¹⁶ Ex-factory price refers to the price at the factory gate, and does not include any other charges, such as delivery or subsequent taxes.

¹⁷ The Levinsohn–Petrin procedure is implemented in this paper using the Stata module "levpet" developed by Petrin, Levinsohn and Poi (2004).

firm *TFP*.¹⁸ Pairwise correlations of the key firm variables are shown in Table A1 in the Appendix.

In this paper, a non-exporter is a firm that had never exported up to and including the reporting year. New exporters are firms that did not export in the previous years in the sample but started exporting in the year of analysis. Their pre-export characteristics can therefore be matched with those of the non-exporting firms. Existing exporters are firms that have export records in previous years and firms that start exporting already in their first year of the sample. This group of firms is dropped in our matching exercises because there are no pre-export characteristics for these firms. However, they are included in the sample for the fixed effects regressions below.

4. Basic patterns

Table A2 in the Appendix reports the key statistics of new exporters, continuing exporters, and non-exporters for the odd years in our sample. For each group of firms, we further separate the sample into domestic and foreign firms. The fraction of domestic exporters (continuing or new exporters) fluctuates between 16% and 24%. This is similar to the U.S., where roughly 20% of plants exported in 1992 (Bernard et al., 2003). Similar to U.S. firms, in China, over 80% of domestic new exporters also sell domestically, and about half of them derive less than 10% of the revenue from foreign sales. Compared to domestic firms, exporters are significantly more prevalent among foreign firms. Between 63% (in 1999) and 72% (in 2004) of foreign firms export.

To compare capital intensity and productivity between exporters (new and continuing exporters included) and non-exporters, we estimate the following specification:

$$\ln S_i = \beta E_i + \{FE\} + \varepsilon_i \tag{1}$$

where S_i is firm *i*'s *TFP* or capital intensity. E_i is a dummy variable indicating the firm's export status. {*FE*} stands for a set of fixed effects, and ε_i is the error term. The percentage differential in S_i between exporters and non-exporting firms can be calculated from the estimated coefficient as $100 \times (\exp(\beta) - 1)$.

In Table 1, Panels A–C report the estimates of Eq. (1) with (log) capital intensity as the dependent variable. In Panel A, real capital stock is measured using the perpetual inventory method and labor is approximated by firm total employment. Column (1) includes E_i and year fixed effects. Column (2) also includes ownership, 4-digit industry, and province fixed effects. To ensure that the change seen is not driven by unobserved firm characteristics, in column (3) we include firm fixed effects (along with year fixed effects). By including firm fixed effects, the correlation between the export status and capital intensity is identified only from firms that switch export status. We find that exporters are on average less capital-intensive than non-exporters. Specifically, in a given year, exporters are about 6% less capital-intensive than non-exporters within a four-digit industry and ownership type (column (2)), and about 5% less capital-intensive within firms (column (3)).

In columns (4) and (5), when we repeat the same analysis for column (3) on domestic and foreign firm samples respectively, we continue to find a negative coefficient on the exporter dummy. It should be noted that the capital intensity gap between exporters and non-exporters is more than double for domestic firms than for foreign firms. By splitting the sample into the pre-WTO period (1999–2001) and the post-WTO period (2002–2007), columns (6) and (7) show that the capital intensity gap increased after China's accession to the WTO, consistent with Figs. 1 and 2. These results contrast sharply with existing studies, which consistently find that exporters are more

capital-intensive (e.g., Bernard and Wagner (1997) for Germany, Bernard and Jensen (2004) for the U.S., Van Biesebroeck (2005) for Sub-Saharan Africa, and De Loecker (2007) for Slovenia).

As a robustness check, we repeat the same set of regressions using alternative measures of firm capital intensity. In Panel B, we measure a firm's real capital stock by the net value of fixed assets deflated by the industry-specific investment price index, whereas in Panel C, it is measured using a firm's total wage bill instead of employment as the denominator. Regardless of how capital intensity is measured, exporters still appear to be less capital-intensive than non-exporters. The estimated capital intensity gap is larger in Panel C. Recent studies show that exporters are larger and tend to employ more skilled workers than non-exporters. This may explain why exporters' capital intensity is even lower when it is calculated using effective labor units.¹⁹ To conserve space, we focus on the results using capital intensity measured by the perpetual inventory method. Using the wage-denominated capital intensity measure yields similar results.

Panel D reports estimates of Eq. (1) with $\ln(TFP)$ as the dependent variable. We find that on average, exporters have higher measured revenue-based *TFP* than non-exporters. This is observed within industries, within a firm, and before and after China's WTO accession, for both domestic and foreign firms. The result that exporters have a higher *TFP* is consistent with most findings in the existing literature.

We now address potential confounding factors in the data. The prevalence of processing exporters in China, who assemble imported intermediates into final products solely for foreign sales, must be considered. According to Kee and Tang (2012), consistently over half of Chinese aggregate exports belongs to processing trade. It is well known that processing exporters in China have lower value added and capital intensity than ordinary (non-processing) exporters. To verify that our results are not driven by the prevalence of processing firms, we repeat the analysis in Table 1 on processing and non-processing exporters. We identify processing and non-processing exporters from the subsample that can be merged with China Customs trade data (see Section 8 for more details on the merged data set). The estimation results are reported in Table A3 in the Appendix. In column (1), only processing exporters are included in the sample and all non-processing exporters are excluded. We still find a strongly negative correlation between export participation and capital intensity within a narrowly defined industry. When only non-processing exporters are considered (column (2)), the negative correlation is even stronger. These results confirm that our main findings are not driven by the prevalence of labor-intensive processing exporters. The lower panel of the table reports the results for firm TFP for processing and non-processing firms. We find that processing exporters are on average less productive than non-exporters, whereas non-processing exporters are more productive. These results confirm previous findings by Dai et al. (2011) and Manova and Yu (2012).

The potentially frequent changes in the export status may affect the results. To partially address this concern, we include three dummies related to export activities in Table A4 in the Appendix: new exporters (a dummy equals 1 for the first year of exporting), continuing exporters (a dummy equals 1 if the firm was exporting in the current year and the previous year), and export stoppers (a dummy equals 1 if the firm exported last year but not this year). We find that new, continuing, and previous exporters are all less capital-intensive (Panel A) and more productive (Panel B) than non-exporters.

The different export intensities of exporters may also affect our results. By using only one dummy variable to capture the export status, we are assuming the same average effects of exporting, regardless of how much the firm exports. In Table A5 in the Appendix, we break down the exporter dummy in Eq. (1) into three dummies to represent

¹⁸ In an earlier version of this paper, we extend the Levinsohn–Petrin procedure by incorporating the firm's export decision into the productivity estimation procedure to control for the export endogeneity problem (Van Biesebroeck, 2005; De Loecker, 2007), instead of estimating productivity using separate production functions for exporters and non-exporters. The results obtained were qualitatively similar.

¹⁹ Harrigan et al. (2012) show that in the US, skill-intensive exporters charge higher prices whereas capital-intensive exporters charge lower prices. This is another example of capital and skill intensity playing different roles in shaping export patterns.

Table 1

Comparing productivity and capital intensity between exporters and non-exporters.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)
	All firms	All firms	All firms	Domestic firms	Foreign firms	Before WTO	After WTO
Panel A: dependent variable	ln(K/L)						
Exporter	-0.192	-0.062	-0.055	-0.069	-0.033	-0.028	-0.067
N	(0.031)	(0.017)	(0.008)	(0.012)	(0.009)	(0.008)	(0.009)
N	1,977,348	1,977,348	1,977,348	1,563,451	413,897	573,694	1,403,654
Panel B: dependent variable	ln(K/L), alternative n	neasure of K					
Exporter	-0.175	-0.029	-0.028	-0.032	-0.022	-0.019	-0.035
	(0.032)***	$(0.005)^{***}$	$(0.007)^{***}$	$(0.008)^{***}$	$(0.008)^{***}$	$(0.009)^{**}$	$(0.010)^{***}$
Ν	1,979,823	1,979,823	1,979,823	1,564,570	415,253	573,694	1,403,654
Panel C: dependent variable	ln(K/L), alternative n	neasure of L					
Exporter	-0.311	-0.143	-0.116	-0.128	-0.098	-0.076	-0.135
	(0.043)***	$(0.026)^{***}$	$(0.021)^{***}$	(0.023)***	(0.025)***	$(0.024)^{***}$	(0.023)***
Ν	1,976,637	1,976,637	1,976,637	1,562,599	414,038	568,121	1,431,480
Panel D: dependent variable	ln(TFP)						
Exporter	0.137	0.087	0.124	0.154	0.034	0.144	0.112
-	(0.039)***	$(0.022)^{***}$	$(0.035)^{***}$	(0.055)***	(0.017)**	$(0.039)^{***}$	(0.032)***
Ν	1,916,347	1,916,347	1,916,347	1,503,658	412,689	543,921	1,372,426
Year FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Industry (4-digit) FE	No	Yes	No	No	No	No	No
Ownership FE	No	Yes	No	No	No	No	No
Firm FE	No	No	Yes	Yes	Yes	Yes	Yes

Notes: this table reports estimation results for Eq. (1) in the text. The exporter dummy equals 1 if a firm is either a new exporter or a continuing exporter. In Panel A, real capital stock (*K*) is measured using the perpetual inventory method, while labor is the firm's total employment. In Panel B, capital stock is the net value of fixed assets deflated by the sector-specific investment deflator, while labor is the firm's total employment. In Panel C, capital stock is measured using the perpetual inventory method, while labor is the firm's total employment. In Panel D, (A) is the firm's total employment of the perpetual inventory method, while labor is the firm's total employment. In Panel D, (A) is the firm's total employment (2003) method. Columns (1)–(3) compare exporters and non-exporters using all firms in the sample; column (4) includes only domestic firms; column (5) includes only foreign firms; columns (6) and (7) split the sample into pre-WTO and post-WTO periods. Standard errors in parentheses are corrected for clustering at the four-digit industry level.

** Indicate significance at the 5% level.

*** Indicate significance at the 1% level.

high intensity exporters (firms with the export/sales ratio larger than 0.9), normal exporters (firms with the export/sales ratio between 0.1 and 0.9), and low intensity exporters (firms with the export/sales ratio smaller than 0.1). Panel A shows significant correlations between each exporter dummy and firm capital intensity. More importantly, the magnitude of the coefficients suggests that a higher export intensity is associated with a lower capital intensity. In Panel B, we find that high intensity exporters are less productive than non-exporters. This is not surprising given that most of the intensive exporters are processing exporters.

5. Changes in capital intensity for new exporters

Table 1 and the corresponding Tables A3 to A5 in the Appendix show the average effects of exporting on capital intensity and *TFP*. In this section, we compare the first difference in capital intensity between non-exporters and new exporters that are ex-ante similar. Existing exporters, who reported positive exports in previous years, are always excluded from this analysis. As a first pass, we follow Lileeva and Trefler (2010) to compare firms within a quartile bin (4 on each dimension) based on firm *TFP* and capital intensity in the previous year. Within each two-digit industry, we assign each firm to one of the 16 bins based on its ex ante *TFP* and capital intensity quartiles. This controls for industry structure and ensures that each bin includes firms from all industries.

In Panel A of Table 2, each cell reports the difference between new exporters and non-exporters in their average capital intensity growth, $\Delta \ln(K/L)_{new\ exporter} - \Delta \ln(K/L)_{non-exporter}$. The first column shows that new exporters in the lowest quartile of *TFP* experience an average 0.047 log-point decline (average across all four entries in the column) in capital intensity after exporting compared to non-exporters within the same *TFP* quartile. The difference is significant at the 1% level.

When we move to higher *TFP* quartiles (moving to the right in each row), we continue to find a post-export decline in capital intensity, but the decline is smaller the higher the *TFP* quartile is. For instance, the average drop in capital intensity (relative to non-exporters) is only 0.028 log points in the highest *TFP* quartile. We will empirically confirm and theoretically explain this negative relationship between firms' ex ante *TFP* and post-export drop in ln (K/L) below.

Table 2

Capital intensity growth: new exporters less non-exporters.

 $\Delta \ln(K/L)_{new\ exporter} - \Delta \ln(K/L)_{non-exporter}$

Panel A		ln(TFP) quartiles before exporting				
		1	2	3	4	
ln(<i>K</i> / <i>L</i>) quartiles before exporting	1 2 3 4	-0.033^{***} -0.045^{***} -0.051^{***} -0.063^{***}	-0.031^{***} -0.039^{***} -0.043^{***} -0.061^{***}	-0.018^{**} -0.026^{***} -0.038^{***} -0.055^{***}	-0.002^{*} -0.019^{**} -0.039^{***} -0.051^{***}	
Panel B		Low	/ TFP	High	n TFP	
Panel B		Low Small firms	r TFP Large firms	High Small firms	n TFP Large firms	

Notes: This table reports the difference in capital intensity growth between new exporters and non-exporters in each bin. All firms are sorted according to their characteristics in the previous year. In Panel A, bins are defined by previous-year $\ln(TFP)$ and $\ln(K/L)$ quartiles. In Panel B, bins are defined by previous-year $\ln(TFP)$, $\ln(K/L)$ and $\ln(sales)$ median. We calculate these bins for each year and each 2-digit industry to ensure each bin covers all years and all industries.

* Indicates significance at the 10% level.

** Indicate significance at the 5% level.

*** Indicate significance at the 1% level.

Table 3	Ta	bl	e	3
---------	----	----	---	---

New exporters' capital intensity ln(K/L)-propensity score matching results.

Bin	TFP	Capital intensity	Size	(1)	(2)	(3)	(4)	(5)
				All firms	Domestic firms	Foreign firms	Alternative K	Alternative L
All bins				-0.051^{***}	-0.054^{***}	-0.042^{***}	-0.027^{**}	-0.069^{***}
1	Low	Low	Small	-0.046^{***}	-0.049^{***}	-0.036***	-0.028^{***}	-0.035***
2	Low	Low	Large	-0.048^{***}	-0.048^{***}	-0.037^{***}	-0.024^{*}	-0.046^{***}
3	High	Low	Small	-0.029^{**}	-0.031^{**}	-0.029	0.001	-0.024^{***}
4	High	Low	Large	-0.032^{**}	-0.041^{**}	-0.027	-0.009	-0.032^{***}
5	Low	High	Small	-0.064^{***}	-0.067^{***}	-0.061^{***}	-0.044^{**}	-0.102^{***}
6	Low	High	Large	-0.061^{***}	-0.066^{***}	-0.053^{***}	-0.029^{*}	-0.113^{***}
7	High	High	Small	-0.063^{***}	-0.068^{***}	-0.051^{***}	-0.043^{***}	-0.096^{***}
8	High	High	Large	-0.059^{***}	-0.062^{***}	-0.61^{***}	-0.039^{***}	-0.103^{***}

Notes: This table reports the estimation results of the impact of exporting on ln(K/L) for new exporters, using the propensity score matching method. The matching is conducted within each bin. These bins (bin 1 to bin 8) are described in Panel B of Table 2. Column (1) includes all firms in the sample; column (2) includes only domestic firms; column (3) includes only foreign firms; column (4) measures capital stock as the net value of fixed assets deflated by the sector-specific investment deflator, while column (5) uses the firm's total wage bill to compute its capital intensity.

* Indicates significance at the 10% level.

** Indicate significance at the 5% level.

*** Indicate significance at the 1% level.

The first row in Panel A shows that among the firms in the lowest quartile of $\ln (K/L)$, the average drop in capital intensity after exporting is 0.021 log-points (average across all four entries in the first row) compared to the non-exporters in the same quartile. When we move down across rows, we find that the relative post-export decline in capital intensity is larger for firms with higher ex ante $\ln (K/L)$. The magnitude of the decline increases to an average of 0.058 log points in the highest $\ln (K/L)$ quartile.

Firm size could be an important factor determining whether firms export, as larger firms are more likely to afford the fixed export costs. Following Lileeva and Trefler (2010), when we construct the bins, we add firm size (measured by ln (sales)) as the third dimension. We assign each firm into one of the eight bins based on its position in *TFP*, capital intensity, and size in its industry.²⁰ Panel B shows no obvious difference in the relative decline in capital intensity between small and large firms. However, the differences between the high and low *TFP* groups are still obvious.

In addition to assigning firms into different bins to compare firms with similar characteristics, we apply the difference-in-difference propensity-score matching estimator (Heckman et al., 1997) to match firms with ex ante similar characteristics. We estimate the propensity score of each firm based on a Probit model with the dependent variable equal to 1 if a firm starts reporting positive exports in the current year. We include a number of pre-export (previous-year) firm characteristics as regressors, namely TFP, wage rate, capital intensity, firm age, firm sales (all in log), 4-digit industry, ownership, and year fixed effects. To minimize the bias due to potentially omitting unobserved characteristics that may affect a firm's decision to export, we do our matching within each of the eight bins as defined in Panel B of Table 2, under the assumption that firms within the same bin are not too different in terms of unobserved characteristics. We use nearest-neighbor matching without replacement.²¹ In total, from all eight bins, 50,856 new exporters were matched to non-exporters in the control group.

Table 3 presents the estimation results of the post-export change in capital intensity using propensity-score matching techniques. In column (1), we find a significant decline in a firm's capital intensity in the year when it starts exporting, compared to non-exporters. This capital intensity gap is observed for all eight combinations of productivity and capital-intensity bins considered in Panel B of Table 2. The gap is larger if the new exporter is ex ante more capital-intensive (bins 5 to 8), consistent with the results in Table 2. In columns (2)-(3), we split the sample into domestic firms and foreign firms for our analysis. We find the same pattern for both types of firms, with a larger drop in capital intensity observed in the domestic firm sample. In column (4), we measure capital stock as the net value of fixed assets deflated by the industry-specific investment price index. The results remain largely robust besides bins 3 and 4. In the last column, we find a larger postexport decline in capital intensity when the wage-denominated measure of capital intensity is used, similar to the findings in Table 1.

To summarize, we have found robust evidence that a Chinese firm becomes less capital-intensive after exporting. We have used various methods – regressions including an exhaustive set of fixed effects, comparisons within bins, and matching within bins – to ensure the robustness of our results. Our findings contrast with the exiting literature that consistently shows higher skill and capital intensity for exporters. Importantly, Tables 2 and 3 show heterogeneous effects on capital intensity across new exporters. The following theoretical section and additional empirical analysis in Sections 7 and 8 will examine these heterogeneous effects.

Given China's comparative advantage in labor-intensive goods, it may not be surprising that exporters in China are less capitalintensive than non-exporters. However, when the pattern is found for both domestic firms and foreign firms within narrowly defined industries and within firms, the standard factor-proportions theory of trade that emphasizes between-sector reallocation of resources cannot provide sufficient explanations. We thus develop a theoretical model to rationalize the findings. Guided by the model, we will further empirically explore the linkages between capital intensity and productivity, both before and after firms' exporting.

6. Theoretical explanation

To explain our empirical findings, we construct a variant of the model by Bernard et al. (2010, 2011) (BRS hereafter). We first briefly discuss the set-up of the BRS model before elaborating in greater detail our extension. Readers are referred to the original paper for details.

Consumers consume a continuum of products with identical prefer-

ences: $U = \left[\int_{0}^{1} C_{s}^{\nu} ds\right]^{\frac{1}{\nu}}$, where $\kappa \equiv 1/(1 - \nu) > 1$ is the elasticity of substitution between products. Within a product, firms produce

²⁰ We no longer construct bins based on quartiles as we would need too many bins (64) for all three dimensions.

²¹ Our results are robust to the use of other matching methods such as local linear regression matching estimator. We choose to do matching without replacement because we have a large number of firms in the control group.

horizontally differentiated varieties, facing their own demand. The consumption index for product *s*, *C*_s, takes the following form:

$$C_{s} = \left[\int_{\omega \in \Omega_{s}} (\lambda_{s}(\omega)c_{s}(\omega))^{\rho} d\omega \right]^{\frac{1}{\rho}}, 0 < \rho < 1,$$
(2)

where $\sigma \equiv 1/(1 - \rho) > 1$ is the elasticity of substitution between varieties within a product. We assume that the elasticity of substitution between varieties within a product is larger than that between products ($\sigma > \kappa > 1$).

Firms are exogenously different along two dimensions, namely firm-specific "ability" and firm-product-specific "consumer" appeals. More specifically, upon paying some fixed (sunk) costs to enter any market, a firm first draws its "ability", $\varphi \in [0, \infty)$, from a distribution $h(\varphi)$. φ is firm-specific and is constant across countries and products. The firm then draws a set of "consumer taste" attributes for each potential product produced, $\lambda_s \in [0, \infty)$, from a distribution $g(\lambda_s)$.²² The set of λ_s is firm-product specific and is constant across countries.²³ To serve market *j*, either domestic or foreign, the firm has to pay an extra fixed cost f_j . In addition to f_j , a multi-product exporter needs to incur product-specific fixed costs, f_{sj} , for each product s sold in market i^{24} An exporter will export product s to market i when its λ_s is sufficiently high and generates enough revenue to cover f_{si} . As in BRS, the more productive exporters have a wider product scope, all else equal, as higher φ generates positive profits for more low- λ_s products.

To rationalize our empirical findings that feature heterogeneous effects on capital intensity, we modify the one-factor BRS model to consider two factors of production — capital and labor. Formally, firms have the following total cost function:

$$TC_{\rm s} = \left[f_{\rm s} + \frac{q_{\rm s}}{\varphi}\right] w^{1-\beta(s)} r^{\beta(s)},\tag{3}$$

where *w* and *r* are the wage rate and the rental rate, respectively. We choose the wage as the numeraire (i.e., w = 1). Notice that the fixed cost to produce a product is assumed to have the same factor shares as the variable costs. $\beta(s)$ represents capital intensity for product *s*. Without loss of generality, we rank product index $s \in [0, 1]$ so that $\beta(0) = 0$, $\beta(1) = 1$, and $\beta'(s) > 0$ (i.e., capital intensity is increasing in product index *s*). Firm profit maximization implies the standard optimal price of a variety exported to country *j* as

$$p_{sj} = \frac{\sigma \tau_j}{\sigma - 1} \frac{r^{\beta(s)}}{\varphi},$$

where τ_j is the iceberg trade cost to country *j*. For simplicity, we assume that τ_j is identical for all products.

Consider two countries: China and destination country *j* (e.g., the U.S.), which is assumed to be more capital-abundant. With trade frictions, factor prices would not be equalized across countries and the wage–rental ratio in country *j* will be higher than that in laborabundant China in equilibrium (i.e., $w_j/r_j > 1/r$). It can then be shown that the relative price of product *s* between country *j* and China, $\tilde{P}_j(s) = P_j(s)/P(s)$, is *decreasing* in capital intensity (i.e., $\tilde{P}_j'(s) < 0$) (see Appendix A.2 for details).²⁵

Given that $P_j(s)$ varies across products, an exporter has a different export portfolio to country *j* compared to the domestic market, even when the set of product attributes (λ_s) is identical for different destinations.²⁶ Consider a firm with φ , the product attribute cutoff $\lambda_s^*(\varphi)$ for product *s*, above which the firm produces *s* for domestic sales, is pinned down by the following zero-profit condition:

$$\pi_{s}(\varphi,\lambda_{s}^{*}(\varphi)) = \frac{R_{s}}{\sigma} \left(r^{-\beta(s)} \rho P(s) \varphi \lambda_{s}^{*}(\varphi) \right)^{\sigma-1} - f_{s} r^{\beta(s)} = 0, \tag{4}$$

where $\pi_s(\varphi, \lambda_s(\varphi))$ represents the firm's profit by selling product *s* domestically; R_s stands for domestic aggregate expenditure spent on product *s*. P(s) is the ideal price index for product $s.^{27}$ Solving Eq. (4) yields the firm-product specific consumer taste cutoff $\lambda_s^*(\varphi)$. Similarly, we can use the zero-profit condition for export sales of product *s* to country *j* to solve for the corresponding product attribute cutoff, $\lambda_{sj}^*(\varphi)$.

Importantly, for product *s*, the firm's product cutoff for exporting to *j* equals

$$\lambda_{sj}^*(\varphi) = \Phi_j(s)\lambda_s^*(\varphi),\tag{5}$$

where $\Phi_j(s) = \tau_j \left(\frac{f_{si}}{f_s} \frac{\hat{p}_j}{\hat{P}_{R_j}}\right)^{\frac{1}{\alpha-1}} \left(\frac{P_j(s)}{\hat{P}(s)}\right)^{-\gamma}$. $\Phi_j(s)$ is increasing in variable (τ_j) and fixed export costs (f_{sj}) , as well as the relative aggregate price index of country j, $\frac{\hat{P}_i}{p}$. $\Phi_j(s)$ is increasing in $\frac{\hat{P}_j}{p}$ because a higher $\frac{\hat{P}_j}{p}$ results in a lower purchasing power of foreign consumers relative to domestic consumers. For the same reason, $\Phi_j(s)$ is decreasing in the total spending of country j, R_j . If countries are symmetric (i.e., $\hat{P} = \hat{P}_j$, $R = R_j$, and $P_j(s) = P(s)$), $\Phi_j(s) = \tau_j \left(\frac{f_{sj}}{f_s}\right)^{\frac{1}{\alpha-1}}$ when international trade costs are higher than domestic trade costs (i.e., $f_{sj} \ge f_s$ and $\tau_j \ge 1$). Given a product attribute, λ_s , $\Phi_j(s) \ge 1$ implies a weakly lower probability of exporting product s, conditional on positive domestic sales. Deviating from the symmetry assumptions, Bernard, Redding, and Schott (2007a) and Lu (2012) postulate the possibility of having $\Phi_j(s) < 1$ and study the resulting implications.²⁸

Suppose $\frac{f_{si}}{f_s}$ and $\frac{R_j}{P_j}$ are invariant across *j*. If country *j* is more capitalabundant than China, $\frac{P_j(s)}{P(s)}$ is decreasing in *s*. Given the assumption that $\sigma > \kappa > 1$, $\frac{\partial \Phi_j(s)}{\partial_s} > 0$. In words, all else being equal, a product attribute that guarantees profitable domestic sales is less likely to generate profitable export sales to *j*, the higher the capital intensity of the product is.

Denote capital cost share for product *s* by $\theta_s = \frac{rk_s}{rk_s+wl_s}$, where k_s and l_s are the total amounts (including fixed cost of production) of capital and labor used to produce *s*.²⁹ Capital intensity of a firm with productivity φ serving only the domestic market is

$$\Theta_d(\varphi) = \int_0^1 \theta_s \widetilde{R}_s(\varphi, \lambda_s) I_s(\lambda_s \ge \lambda_s^*(\varphi)) ds = \int_0^1 \left[\int_{\lambda_s^*(\varphi)}^\infty \theta_s \widetilde{R}_s(\varphi, \lambda_s) g(\lambda_s) d\lambda_s \right] ds,$$

²⁶ In BRS, there is a Poisson probability for the firm to draw firm-specific productivity term, and another Poisson probability that the firm draws a new consumer taste for a product. It is theoretically possible that a firm gets hit by a positive productivity shock and decides to export, while its product-specific consumer taste shocks do not change. Moreover, we follow BRS to assume that the distribution of abilities and product attributes are independent of one another.

²⁷ Specifically, consumers' utility maximization yields $R_s = \left[P(s)^{\frac{pr}{1-p}} / \int_0^1 P(k)^{\frac{pr}{1-p}} dk\right]$, where R is total expenditure of the economy; $P(s) = \left[\int_{w \in \Omega_s} P(s, w)^{1-\sigma} dw\right]^{\frac{1}{1-\sigma}}$.

²² We could have modeled the firm-product specific draw as productivity, but it will not change any of the theoretical results. The reason is that with CES demand and monopolistic competition, firm-product productivity and firm-product demand shifter have exactly the same effects on firm revenue and profits in equilibrium, as is also argued by BRS.

 $^{^{23}}$ Due to the law of large number, as long as we assume that firms draw product-destination specific attribute, λ_{sj} from the same distribution, our main theoretical results will continue to hold.

 $^{^{24}}$ Think of f_{sj} as R&D expenditure required to produce a blue print for the product or the overhead costs to manage the product-specific sales team.

²⁵ A similar point has been made by Lu (2012) who rationalizes why Chinese exporters are less productive than domestic producers in sufficiently labor-intensive sectors.

²⁸ In particular, Lu (2012) finds that in labor-intensive sectors, Chinese exporters are on average less productive than non-exporters. Based on an extension of Bernard et al. (2007b), she rationalizes the findings by postulating that if the domestic is more competitive than the foreign market, the domestic production cutoff can be lower than the export participation cutoff.

²⁹ E.g. $rk_s = rk_s^p + rk_s^f$, where k_s^p stands for the level of capital used for producing goods, while k_s^f is the corresponding amount to cover the fixed cost of production, such as developing a blue print of the product.

where subscript *d* denotes "domestic"; $I_s(\lambda_s \ge \lambda_s^*(\varphi))$ is an indicator function, which equals 1 if $\lambda_s \ge \lambda_s^*(\varphi)$. $\tilde{R}_s(\varphi, \lambda_s) = \frac{R_s(\varphi, \lambda_s)}{R(\varphi)}$ is the ratio of the firm's domestic sales of product *s* to its total domestic sales. $g(\lambda_s)$ is the stationary distribution of consumer tastes, which is discussed in detail in BRS. The last equality holds because of the law of large number: the firm's capital intensity is equal to the *expected* weighted average across all products' capital intensity.

Conditional on export participation in market *j*, we can similarly derive the firm's capital intensity of the basket of goods exported to *j* as

$$\Theta_{j}(\varphi) = \int_{0}^{1} \theta_{s} \widetilde{R}_{sj}(\varphi, \lambda_{s}) I_{s} \Big(\lambda_{s} \geq \Phi_{j}(s) \lambda_{s}^{*}(\varphi) \Big) ds = \int_{0}^{1} \left[\int_{\Phi_{j}(s) \lambda_{s}^{*}(\varphi)}^{\infty} \theta_{s} \widetilde{R}_{sj}(\varphi, \lambda_{s}) g(\lambda_{s}) d\lambda_{s} \right] ds,$$

where $\widetilde{R}_{sj}(\varphi, \lambda_s) = \frac{R_{sj}(\varphi, \lambda_s)}{R_j(\varphi)}$ is the ratio of the firm's product *s* sales in foreign market *j* to the its total sales there. We assume that θ_s is identical for product *s* in different markets.³⁰ A firm selling both at home and country *j* thus has the following capital intensity:

$$\Theta_{d+j}(\varphi) = \left(1 - \frac{R_j(\varphi)}{R(\varphi) + R_j(\varphi)}\right) \Theta_d(\varphi) + \frac{R_j(\varphi)}{R(\varphi) + R_j(\varphi)} \Theta_j(\varphi).$$
(6)

As in BRS, given a continuum of products, the law of large number implies that a firm's exporting status is entirely determined by firm productivity, φ , and an overall fixed cost for exporting to country *j*, *f*_i. Given φ and $\lambda_{sj}^*(\varphi)$, firm expected profit from serving market *j* is

$$\pi_j(\varphi) = \int_0^1 \left[\int_{\lambda_{s_j}^*(\varphi)}^\infty \pi_{s_j}(\varphi, \lambda_s) g(\lambda_s) d\lambda_s \right] ds - f_j$$

where f_j is measured in labor in BRS, but is measured in the domestic consumption bundle here.

Consider a firm that initially serves only the domestic market at period t (i.e., $\pi_{j,t}(\varphi) < f_i$). Suppose a shock hits the firm and lowers its fixed export cost to f_j' such that $\pi_{j,t+1}(\varphi) > f_j'$. The firm starts exporting to country j at period t + 1.³¹ For the moment, consider sufficiently high trade costs so that all product cutoffs for foreign sales are higher than the corresponding ones for domestic sales, $\Phi_j(s) > 1$ and thus $\lambda_{sj}^*(\varphi) > \lambda_s^*(\varphi) \forall s$.³² Since $\frac{\partial \lambda(s)}{\partial s} > 0$, given the same distribution function $g(\lambda_s) \forall_s$, the firm is more likely to have a λ_s that is higher than both $\lambda_s^*(\varphi)$ and $\lambda_{sj}^*(\varphi)$ for labor-intensive (low s) products. In other words, the firm is less likely to have λ_s that justifies exports of capital-intensive (high s) products, even though the firm could be already selling the same product at home. Given a continuum of products, the (weighted) average capital intensity of the products sold domestically is the same before and after exporting (i.e., $\Theta_{d,t}(\varphi) = \Theta_{d,t+1}(\varphi)$). The (weighted) average capital intensity for the export bundle will be lower (i.e., $\Theta_{j,t+1}(\varphi) = \Theta_{d,t+1}(\varphi)$). In sum, we have the following proposition: **Proposition 1.** A firm's capital intensity $\Theta(\varphi)$ after switching from nonexporting at period t to exporting to a capital-abundant country j at period t + 1 satisfies the following inequality:

$$\Theta_{j,t+1}(\varphi) < \Theta_{t+1}(\varphi) < \Theta_{d,t+1}(\varphi) = \Theta_t(\varphi),$$

where $\Theta_t(\varphi)$ and $\Theta_{t+1}(\varphi)$ are the capital intensities of the firm before and after exporting. $\Theta_{d,t+1}(\varphi)$ and $\Theta_{j,t+1}(\varphi)$ are the capital intensities of the domestic and foreign baskets of products after exporting.

Given the definition of firm capital intensity in Eq. (6), Proposition 1 provides an explanation to our empirical findings that firms become less capital-intensive after exporting to a capital-abundant country. Notice that Proposition 1 does not require an assumption that $\lambda_{sj}^*(\varphi) > \lambda_s^*(\varphi)$ $\forall s$. For it to hold, what is needed is simply $\frac{\partial \Phi_j(s)}{\partial_s} > 0.^{33}$ In Appendix A.2, we show that as long as there are some s with $\lambda_{sj}^*(\varphi) > \lambda_s^*(\varphi), \frac{\partial \Phi_j(s)}{\partial s} > 0$ o suffices to guarantee a decline in capital intensity of a new exporter serving j.

Furthermore, keeping all other aspects of the destination countries identical, there must exist a product $\tilde{s} \in (0, 1)$ such that $\Phi(\tilde{s}) = \Phi_j(\tilde{s})$, $\Phi_j(s)/\Phi(s) < 1 \forall s < \tilde{s}$ and $\Phi_j(s)/\Phi(s) > 1 \forall s > \tilde{s}$. As such, a new exporter can adjust its product scope at two margins to end up with a lower capital intensity. The first margin is the intensive margin – firms will increase sales of the existing labor-intensive products after exporting. The second margin is the extensive margin – firms will also add products that have λ_s satisfying $\lambda_s^*(\varphi) > \lambda_s > \Phi_j(s) \lambda_s^*(\varphi)$ to the exported product portfolio, but not to the domestic product portfolio.³⁴ Notice that this situation is more likely to happen for labor-intensive products as $\frac{\partial \Phi_j(s)}{\partial_s} > 0$. Adjustments at either margin imply an increase in the share of labor-intensive products in the firm's product portfolio, contributing to a decline in its capital intensity.

Based on the intuition behind Proposition 1, we have the following corollary about the relation between destinations' capital abundance and exporters' capital intensity.

Corollary 1. Between two ex ante identical firms, the one that starts exporting to a country that is more capital-abundant but identical otherwise will experience a larger decline in capital intensity.

Proof. See Appendix A.2.

The main idea behind Corollary 1 is that in a more capital-abundant country *j*, the product cutoff schedule, $\Phi_i(s)$, is stepper.

Two remarks are in order before we move on to discussing the heterogeneous effects. First, to fix idea, our model focuses on exporting to a single country. It can be extended to consider a firm's exporting to multiple countries.³⁵ As long as the destination countries are on average more capital-abundant than the exporting country (China in this case), an extension to a multi-country setting will not change the main theoretical result. Our model does permit the situation that fixed cost shocks trigger exporting to the more

$$\begin{split} \Theta_X(\varphi) &= \bigg(1 - \frac{R_X(\varphi)}{R(\varphi) + R_X(\varphi)}\bigg) \Theta_d(\varphi) + \frac{1}{R(\varphi) + R_X(\varphi)} \sum_{j=1}^N (R_j(\varphi) \Theta_j(\varphi)), \\ \text{where } R_X(\varphi) &= \sum_{i=1}^N R_j(\varphi). \end{split}$$

³⁰ Recent literature has shown that within the same narrowly defined product category, product quality is higher for exports to richer destinations (Bastos and Silva, 2010; Manova and Zhang, 2012a, 2012b). If product quality is positively corrected with skill and capital intensity, as is shown by Verhogeen (2008), adjusting for the effect of quality on factor intensity should strengthen our findings of lower capital intensity among exporters.

³¹ A firm can also switches from non-exporting to exporting after receiving a favorable shock to productivity, φ . With a few mild assumptions, our main theoretical results will go through. Since our empirical analysis has focused on comparing exporters and non-exported with similar ex-ante characteristics, including productivity, we choose to focus on the case of fixed-cost shocks to more closely link our theory to the empirical results.

³² Bernard et al. (2007b) make a similar assumption – the productivity cutoffs to export are higher in both capital- and labor-intensive sectors.

³³ In fact, we can do the same exercise as Lu (2012) and assume that there exists $\overline{s}(\varphi) < 1$ such that $\lambda_{sj}^*(\varphi) \leq \lambda_s^*(\varphi) \forall_s \leq \overline{s}(\varphi)$, and $\lambda_{sj}^*(\varphi) > \lambda_s^*(\varphi)$ otherwise.

³⁴ For simplicity, our model does not analyze product dropping due to the generalequilibrium competitive effects of trade liberalization (e.g., Bernard et al., 2011). Another way to analyze product dropping is to extend our model to incorporate firms' capacity constraints or cannibalization effects from introducing new products (e.g., Dhingra, forthcoming).

 $^{^{35}}$ For instance, the capital intensity of a firm exporting to *N* different countries will be

labor-abundant countries. We will examine the differential effect of exporting to destinations with varying capital abundance in the empirical analysis below.

Second, our model focuses on product churning as the main driver of changes in a firm's factor intensity. By no means this is the only channel through which exporting matters. Firms may still invest in new capital or capital-intensive activities, as shown by existing studies (e.g., Aw et al., 2011; Bustos, 2011). It is worth noting that our empirical results so far show a net effect of exporting on capital intensity, which imply that the standard investment effects proposed in the literature is dominated by the product-churning effects on average in China. Importantly, our model is general enough to incorporate the investment decisions considered by those studies for an examination of the gross effects of product churning on a firm's factor intensity.

Besides rationalizing the main findings, our model yields additional predictions about the heterogeneous responses across firms. First, it relates the firm's ex ante productivity to its post-export change in capital intensity. In particular, our model shows that $\lambda_s^*(\varphi)$ is decreasing in φ for all *s*, which implies that the more productive firms have lower product cutoffs for both domestic and foreign sales and can afford to sell a wider range of products in any given market. Thus, the ex ante more productive firms are expected to have less product churning and thus a smaller decline in capital intensity after exporting. Using the same firm-level data, we will examine the following proposition.

Proposition 2. An ex ante more productive firm experiences a smaller decline in capital intensity after exporting. Formally,

$$\frac{\Theta_{t+1}(\varphi)}{\Theta_t(\varphi)} \!\! < \!\! \frac{\Theta_{t+1}\left(\varphi'\right)}{\Theta_t(\varphi')} \! < \!\! 1 \hspace{0.1cm} \textit{if} \hspace{0.1cm} \varphi' \!\! > \!\! \varphi$$

6.1. Revenue-based productivity estimates

Our empirical results show that firms become more productive after exporting. As discussed in the literature review, there are many reasons for that to happen. BRS has shown that trade liberalization would lead to more intense competition in the domestic product market, inducing firms to reduce their product scope and allocate resources toward the best performing products. Building on their model, our model naturally delivers similar predictions. Therefore we do not aim to replicate the theoretical results from BRS, but instead develop a testable hypothesis that relates the change in a firm's capital intensity due to product selection to its productivity after exporting.

The revenue-based *TFP* measure associated with domestic sales of product *s* by a firm with φ is

$$\mu_s = \frac{R_s(\varphi, \lambda_s)}{x_s(\varphi, \lambda_s)},\tag{7}$$

where $x_s(\varphi, \lambda_s) = \Gamma_s l(\varphi, \lambda_s)^{1 - \beta(s)} k(\varphi, \lambda_s)^{\beta(s)}$ is the associated input bundle, and Γ_s is a sector-specific constant that delivers a cost function equal to Eq. (3). By expressing the quantity produced as $q_s(\varphi, \lambda_s) = \varphi(x_s(\varphi, \lambda_s) - f_s)$ and revenue as $R_s(\varphi, \lambda_s) = \frac{r^{\beta(s)}}{\rho\varphi}q_s(\varphi, \lambda_s)$, we can rewrite Eq. (7) as:

$$\mu_{s} = \frac{r^{\beta(s)}}{\rho} \left(1 - \frac{f_{s}}{x_{s}(\varphi, \lambda_{s})} \right)$$

 $x_s(\varphi, \lambda_s)$ and thus μ_s are increasing in λ_s and φ . The intuition is that a firm with a higher φ or λ_s produces more and can spread the fixed cost of production f_s over a larger volume of output. Similarly, the

product-specific measured *TFP* corresponding to foreign sales in country *j* is

$$\mu_{sj} = \frac{\tau_j r^{\beta(s)}}{\rho} \left(1 - \frac{f_{sj}}{x_{sj}(\varphi, \lambda_s)} \right)$$

By definition, the measured revenue-based *TFP* of an exporter that sells both domestically and in country *j* is the weighted average of μ 's, with weights equal to the revenue shares of the products:

$$\begin{split} \widehat{TFP}_{j}(\varphi) &= \int_{0}^{1} \mu_{s} \int_{\lambda_{s}^{*}(\varphi)}^{\infty} \frac{R_{s}(\varphi, \lambda_{s})}{R(\varphi) + R_{j}(\varphi)} g(\lambda_{s}) d\lambda_{s} ds \\ &+ \int_{0}^{1} \mu_{sj} \int_{-\phi_{j}(s)\lambda_{s}^{*}(\varphi)}^{\infty} \frac{R_{s}(\varphi, \lambda_{s})}{R(\varphi) + R_{j}(\varphi)} g(\lambda_{s}) d\lambda_{s} ds, \end{split}$$
(8)

where the first term corresponds to *TFP* measured based on domestic sales and the second term corresponds to *TFP* measured based on foreign sales. While a positive shock on a firm's φ will increase $\widehat{TFP}_j(\varphi)$, our model focuses on shocks that lower a firm's fixed export costs. This modeling approach is consistent with our empirical strategy that matches and compares firms with similar ex ante *TFP* and other characteristics. The identification assumption of our matching exercises is that in the absence of within-firm reallocation of resources, firms that are similar over a wide range of observables are expected to remain similar ex post in the observables. That said, we cannot rule out positive productivity shocks as a trigger to export participation.

How can product churning after exporting affect firms' measured *TFP*? Consider a model with symmetric countries (i.e., identical country size and factor endowment). Costly trade implies a selection of better performing products into a firm's export basket. This can be shown by $\Phi_j(s) > 1$ in our model under the symmetry assumption. The average λ_s (either simple or sales-weighted) is then higher for the products exported than those sold domestically. This pattern of product selection has been empirically verified by Arkolakis and Muendler (2011) and Manova and Zhang (2012a) and theoretically shown to contribute to a higher firm measured *TFP* by BRS (2010), among others.

However, when countries are asymmetric in size and/or factor endowment, the contribution of product switching to firm measured *TFP* becomes less straightforward. In particular, $\Phi_j(s)$ can be higher or lower than 1 when the relative price index is not constant across *s*. Complicating the analysis is that the revenue-based "productivity" of product *s*, μ_{sj} , depends positively on λ_s but negatively on f_{sj} . Specifically, higher fixed cost for exporting than domestic sales ($f_{sj} > f_s$) implies that on one hand, product selection pushes up firm measured *TFP*, but on the other hand forces μ_{sj} lower than μ_s for some products, which then reduces firm measured *TFP*. However, given f_s , f_{sj} , φ and λ_s we can show that $\mu_{sj} > \mu_s$ if ³⁶

$$\frac{f_{sj}}{f_s} < \left(\frac{P_j(s)}{P(s)}\right)^{\gamma} \Psi_j, \tag{9}$$

where $\Psi_j = \frac{R_j/\hat{P}_j}{R/\hat{P}}$ is constant across *s*.

Considering constant $\frac{f_{sj}}{f_s}$ across *s*, since $\tilde{P}'(s) < 0$ and $\gamma \equiv \frac{\sigma(1-\nu)-1}{(\sigma-1)(1-\nu)} > 0$, the right hand side of the inequality is decreasing in *s*. That is, this inequality is more likely to hold for labor-intensive products, all else being equal. Thus, the more specialized a firm is in labor-intensive products after exporting, the more weight it will have on exported products that have $\mu_{sj} > \mu_s$, which will in

$$\begin{array}{rl} {}_{36} \hspace{0.2cm} \mu_{sj} \! > \! \mu_{s} \! \Rightarrow \! \frac{f_{ij}}{x_{ij}(\varphi,k_{s})} \! < \! \frac{f_{ij}}{x_{ij}(\varphi,k_{s})} \! < \! \frac{f_{ij}}{\varphi} \! \\ {} \Rightarrow \! \frac{f_{ij}}{\frac{r_{ij}(\varphi,k_{s})}{\varphi} \! + \! f_{ij}} \! < \! \frac{q_{ij}(\varphi,k_{s})}{\varphi} \! \\ {} \Rightarrow \! \frac{R_{ij}(\varphi,k_{s})}{R_{ij}(\varphi,k_{s})} \! > \! \frac{f_{ij}}{f_{s}} \! . \end{array}$$

turn result in a higher measured *TFP*. ³⁷ Intuitively, given constant fixed export costs across products, an exporter is able to derive higher profits from foreign sales than domestic sales for labor-intensive products, all else equal. Higher profitability is then translated into higher revenue-based *TFP* at the product level. We summarize the relation between the change in capital intensity and measured *TFP* of the firm in the following proposition:

Proposition 3. Firms that experience a bigger decline in capital intensity after exporting have a larger increase in measured revenue-based TFP.

7. Evidence on heterogeneous changes in capital intensity

Tables 1 to 3 have already shown robust evidence supporting Proposition 1. We now verify Proposition 2 by examining whether a firm's ex ante *TFP* can affect the change in capital intensity after exporting. To this end, we estimate the following equation:

$$\Delta \ln (K/L)_i - \Delta \ln (K/L)_i^{matched} = \delta \ln \left(TFP_{i,t-1} \right) + X_i \gamma + \{FE\} + \zeta_i, \quad (10)$$

where $\Delta \ln(K/L)_i$ is the change in firm *i*'s capital intensity from year t - 1 to *t* when it starts exporting. $\Delta \ln(K/L)_i^{matched}$ is the change in the capital intensity of *i*'s matched non-exporter over the same period. The match is determined by the propensity scores of the firms estimated for the analysis in Table 3. The regressor of interest is firm *i*'s ex ante *TFP*, $\ln(TFP_{i,t-1})$. X_i is a vector of firm *i*'s characteristics in year t - 1, which include the firm's wage rate, capital intensity, and age (all in logs). {*FE*} includes ownership, 4-digit industry, and province fixed effects. Proposition 2 predicts that $\delta > 0$.

The estimates of Eq. (10) are reported in Table 4. The positive coefficient on $\ln(TFP_{t-1})$ in column (1) shows that higher firm *TFP* before exporting is on average associated with a smaller decline in capital intensity after exporting, relative to the matched non-exporters. These results support Proposition 2. The positive and significant coefficient on ln (wage rate) also provides consistent results, if the more productive firm pay higher wages.

We also find that the ex ante more capital-intensive firms are on average associated with a larger decline in capital intensity after exporting. Our model specifies that two non-exporting firms with the same productivity should have the same capital intensity. It is thus silent about the relation between the level of firm capital intensity and export participation. But suppose a capital-intensive firm starts exporting, a wide range of the products that it sells in the domestic market, which tends to be capital-intensive, cannot be exported profitably to a capitalabundant country. As such, in a labor-abundant country, a more capitalintensive firm will tend to experience a larger decline in capital intensity after exporting to a capital-abundant country. In columns (2) and (3), we find strong evidence confirming the baseline results using both the domestic and foreign firm samples.

Next, we explore the relationship between the change in capital intensity in the first year of exporting and the gain in measured *TFP* to shed light on the "core competency" hypothesis, according to Proposition 3. We regress the change in measured *TFP* of new exporters relative to the corresponding change of the matched non-exporters. Table 5 reports the results. Column (1) shows a negative coefficient on the change in capital intensity after exporting, controlling for 4-digit industry, ownership, and year fixed effects, as well as a number of key firm attributes. This result suggests that firms with a deeper specialization in labor-intensive products (i.e., a decline in capital intensity) experience a larger increase in measured *TFP*, supporting Proposition 3. This correlation remains robust for both domestic (column (2)) and foreign exporters (column (3)).

Table 4

Ex ante productivity and ex post capital intensity.

Dependent variable = $\Delta \ln(K/L)_{new \ exporter} - \Delta \ln(K/L)_{matched \ non-exporter}$

	(1)	(2)	(3)
	All new exporters	Domestic new exporters only	Foreign new exporters only
$ln(TFP_{t-1})$	0.096	0.132	0.021
	(0.035)***	(0.049)***	$(0.008)^{**}$
$\ln(K_{t-1}/L_{t-1})$	-0.298	-0.345	-0.276
	(0.074)***	(0.092)***	(0.077)***
$ln(wage rate_{t-1})$	0.188	0.193	0.198
	(0.048)***	(0.054)***	$(0.088)^{**}$
$ln(age_{t-1})$	0.029	0.035	0.027
	(0.011)***	(0.013)***	(0.011)**
Ownership FE	Yes	No	No
Industry (4-digit) FE	Yes	Yes	Yes
Province FE	Yes	Yes	Yes
Ν	49,742	33,409	16,333

Notes: All regressors are lagged by one year. New exporters and their matched nonexporters are matched using the propensity score matching method, which is done within their own bins. Standard errors in parentheses are corrected for clustering at the four-digit industry level.

** Indicate significance at the 5% level.

*** Indicate significance at the 1% level.

8. Evidence on within-firm product switching

In the remainder of the paper, we use transaction-level (firm-product-year) trade data to verify our main theoretical predictions. We first merge the NBS above-scale manufacturing firm data with the customs transaction-level trade data as discussed in Section 3. We use various methods to merge the two data sets, merging by firm name, address, and manager names. The summary statistics of the merged data set are reported in Appendix Table A6. About one-third of the exporters in the trade data set can be merged with the NBS data set. These merged firms account for 37 to 49% (depending on the year) of the values of aggregate Chinese exports. A conservative estimate shows that over 20% of Chinese exports were intermediated by trading companies (Ahn et al., 2011; Tang and Zhang, 2012). It should be noted that trading companies are considered service providers, which are included in the trade data but not in the NBS industrial firm data. A large fraction of the unmerged firms in our sample are thus trading companies.

Tabl	e 5	

Determinants of the change in firm TFP. Dependent variable = $\Delta \ln(TFP)_{new exporter} - \Delta \ln(TFP)_{matched non-exporter}$

	-	-	
	(1)	(2)	(3)
	All new exporters	Domestic new exporters only	Foreign new exporters only
$\Delta \ln(K/L)_{t-1,t}$	-0.049	-0.055	-0.022
	(0.008)***	(0.019)***	(0.010)**
$\ln(TFP)_{t-1}$	0.121	0.135	0.119
	(0.025)***	(0.030)***	(0.031)***
$ln(wage rate)_{t-1}$	0.087	0.093	0.076
	(0.001)***	(0.002)***	(0.002)***
$ln(age)_{t-1}$	-0.076	-0.086	-0.078
	(0.029)***	(0.035)**	(0.037)**
Industry (4-digit) FE	Yes	Yes	Yes
Onwership FE	Yes	No	No
Province FE	Yes	Yes	Yes
Ν	50,019	33,637	16,382

Notes: All regressors are lagged by one year, besides $\triangle \ln(K/L)_{t-1,t}$ which is defined as the first difference in capital intensity from year t-1 to t. Standard errors in parentheses are corrected for clustering at the four-digit industry level.

** Indicate significance at the 5% level.

*** Indicate significance at the 1% level.

³⁷ That said, it is possible that $\mu_{sj} < \mu_s$ even for the most labor-intensive products exported to capital-abundant countries. This would be the case if f_{sj} is significantly higher than f_{s} or the destination country is sufficiently small (low R_i) or remote (high \hat{P}_i).

Table 6

Product switching of new exporters (customs transaction-level data).

	(1)	(2)	(3)	(4)	(5)
	Number of new exporters	Number of new exporters that survived to next year	Total (average) number of products added next year	Total (average) number of products dropped next year	Total (average) number of continuing products
2001	15,928	13,187	134,059 (10.17)	56,389 (4.28)	63,929 (4.85)
2002	21,383	18,410	176,066 (9.56)	82,096 (4.46)	98,364 (5.34)
2003	27,107	22,941	229,762 (10.02)	127,959 (5.58)	125,753 (5.48)
2004	37,646	31,583	322,921 (10.22)	207,112 (6.56)	161,901 (5.13)
2005	40,024	33,552	311,839 (9.29)	265,860 (7.92)	166,894 (4.97)
Average	28,418	23,935	234,929 (9.85)	147,883 (5.76)	123,368 (5.15)

Notes: A product is defined as an HS6 category. In columns (3) to (5), average number of products equals total number of products divided by the number of new exporters that survived to next year in column (2). We do not include 2006 new exporters because we have no information on their survival and number of products added and dropped in 2007.

Using the merged data set, we compute capital intensity of each HS 6-digit product. The computation procedures, similar to the method used by Bernard et al. (2010), are discussed in Appendix A.3. Appendix Table A8 reports the measured capital intensity by broad sector. Similar to the findings by Bernard et al. (2010) for the U.S., we find a large variation in capital intensity within a sector. For instance, the mean capital intensity of the "textiles and textile articles" sector is about 68 thousand yuan per worker, while the standard deviation across HS6 products within the same sector is about 55 thousand. There are many HS6 product categories within a sector. The number of HS6 product categories ranges from 9 (Works of art) to 818 (Textile and textile articles), suggesting that firms in the same sector have a wide range of products with vastly different capital intensities to choose from.³⁸ Using the transaction-level data, we find that exporters actively add and drop products over time. Table 6 shows that from 2002 to 2006, new exporters on average added about ten products, dropped six products, and continued only five products after the first year of exporting.³⁹ This active within-firm extensive margin of trade can play an important role in affecting factor intensity and measured productivity after export participation.

Using the merged data set and capital intensity measures at the HS 6-digit level, we compare the (average) capital intensity of the products that were newly added, dropped, and continued from the previous year at the firm level. In the year immediately after a firm starts exporting, we assign its exported products (HS6) into three categories: the newly added, continued, and dropped products. For each of the new exporters in the year right after the first year of exporting, we compute three sales-weighted averages of capital intensities across products, one for each category. Using this data set, we estimate the following specification:

 $\ln (K/L)_{ik} = \eta_0 + \eta_1 \ new_product_{ik} + \eta_2 \ dropped_product_{ik} + e_i, \quad (11)$

where $\ln(K/L)_{ik}$ is firm *i*'s sales-weighted average capital intensity of product category $k, k \in \{\text{new products, dropped products, continued products}.$ *new_product*_{ik} and*dropped_product*_{ik} are dummy variables to indicate that firm*i*added new products and dropped old products respectively. More specifically,*new_product*_{ik} equals 1 if firm*i* $added new products in the year after the first year of exporting, with the corresponding dependent variable <math>\ln(K/L)_{ik}$ measuring the average capital intensity of the these new products added. Similarly, *dropped_product*_{ik} equals 1 if firm *i* dropped some old products, with the corresponding $\ln(K/L)_{ik}$ measuring the average capital intensity of those dropped products. The omitted reference group is the category that firm *i* continuous-ly exported some existing products. η_0 is a constant and e_i is the error term. Our model predicts that new products are less capital-intensive than the continued products, while dropped products are more capital-intensive. Thus, $\eta_1 < 0$ and $\eta_2 > 0$.

As shown in Table 7, the estimated coefficient on the new-product dummy is negative and significant using the pooled sample, while the dropped product dummy is positive and significant. More specifically, the new products are about 5% less capital-intensive than the continuously exported products and the dropped products are about 2% more capital-intensive.

We conduct several robustness checks in columns (2)-(6). In column (2), we use the deflator approach instead of the perpetual inventory approach. The capital stock is measured as the net value of fixed assets deflated by the sector-specific investment deflator. Again we find statistically significant results for both the new product dummy and the dropped product dummy. As discussed in Appendix A.3, our preferred method of calculating product capital intensity is to use the weighted average of the capital intensity of all firms exporting that product. However, by using the weighted average, our measure of product capital intensity may be dominated by a few large exporters that export multiple products. To reduce such bias, we use the median capital intensity in column (3) and the results remain qualitatively similar. Our results are insensitive to the exclusion of intermediaries (column (4)), and remain robust when we include only ordinary exporters (column (5)) or processing exporters (column (6)). These findings address the concern that our results are driven by the predominance of processing exporters in China.

Our model predicts that exporting to a more capital-abundant country should be associated with a larger decline in firm capital intensity. To examine this hypothesis, we split the sample into two, a group of exporters primarily serving the capital-abundant countries and a group of exporters primarily serving the labor-abundant countries. Our country capital abundance data come from Antweiler and Trefler (2002). A country is considered capital-abundant (labor abundant) if its relative capital endowment is higher (lower) than the median value in the Antweiler and Trefler (2002) sample. For multi-country exporters, we classify firms based on their largest export destination. We run the same regression over the two groups. As shown in columns

³⁸ Our product-level measure of capital intensity may introduce a selection or aggregation bias. The production choice of a firm depends on its productivity. More productive firms may produce more products. Since China is labor-abundant, firms will add products in decreasing order of capital intensity. As a result, our measure of the capital intensity of products may be biased upward, since our measure is based on the relatively more productive firms that sell more of these products than the less productive firms. By using the median capital intensity, we find that the empirical results remain qualitatively similar.

³⁹ We do not observe the product mix of new exporters in the transaction-level trade data before they start exporting. Hence we cannot measure the decline in firms' capital intensity in the first year of exporting. We can only measure the changes between the first year of exporting and subsequent years. An implicit assumption behind this empirical approach is that there are adjustment costs preventing firms from reaching the optimal product mix within the first year of exporting.

Table 7

Capital intensity of new products and dropped products.

Dependent variable: ln(K/	L)							
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
	All New exporters	Alternative measure of K	Alternative construction of product <i>K/L</i>	Excluding intermediaries	Ordinary exporters only	Processing exporters only	Capital abundant destinations	Labor abundant destinations
New product category dummy Dropped product category dummy Year fixed effects N	-0.048 (0.012)*** 0.023 (0.005)*** Yes 326,784	-0.019 (0.005)*** 0.016 (0.004)*** Yes 326,784	-0.025 (0.005)**** 0.009 (0.004)*** Yes 326,784	-0.049 (0.014)*** 0.021 (0.007)*** Yes 281,035	-0.050 (0.015) ^{***} 0.024 (0.008) ^{***} Yes 252,887	-0.045 (0.013) ^{***} 0.013 (0.005) ^{***} Yes 73,897	-0.049 (0.013)*** 0.022 (0.007)*** Yes 262,733	-0.042 (0.014)*** 0.023 (0.007)*** Yes 41,536

Notes: This table reports the results of regressions of capital intensity on the new product category dummy and the dropped product category dummy. The omitted group is the continuing product category. Column (1) uses the capital stock measure computed using the perpetual inventory method (our benchmark method) to calculate product capital intensity. In column (2), capital stock is the net value of fixed assets deflated by the sector-specific investment deflator. In column (3), we use the median capital intensity of all firms exporting the same product to calculate product capital intensity. Column (4) excludes all trade intermediaries. Columns (5) and (6) use the subsamples of ordinary (non-processing) and processing new exporters. Column (7) and (8) use the subsamples of the exports to capital-abundant countries and labor-abundant countries. The classification of capital abundance is based on Antweiler and Trefler (2002). Robust standard errors are reported in parentheses.

** Indicate significance at the 5% level.

*** Indicate significance at the 1% level.

(7) and (8), products added by exporters serving capital-abundant destinations tend to be less capital-intensive than those added by exporters serving labor-abundant destinations. There is little difference in the factor intensity of the dropped products between the two groups of exporters.

9. Concluding remarks

This paper studies how a firm's specialization in its core products after exporting affects its factor intensity and productivity. Using panel data for China's manufacturing firms over the 1998–2007 period and several empirical methods, we find that a firm becomes less capitalintensive but more productive after exporting. For both domestic firms and foreign firms, this fact is established within firms, within a narrowly defined industry, and within a group of firms with similar ex ante characteristics.

As our findings on post-export capital intensity contrast sharply with the existing findings in the literature, we develop a variant of the model by Bernard et al. (2010, 2011) to consider firms producing multiple products with varying capital intensity. Our model predicts that firms in labor-abundant countries specialize in their core competencies by allocating more resources to produce labor-intensive products once they start exporting. We discuss how this within-firm reallocation of resources is related to firm measured productivity after exporting. Firm ex ante productivity is associated with a smaller decline in capital intensity after exporting, while a sharper post-export decline in capital intensity is associated with a larger increase in measured total factor productivity. Using transaction-level trade data, we find that during our sample period, new exporters in China add new products that are more labor-intensive than the existing exported products in subsequent years and drop those that are less labor-intensive. These product-churning patterns tend to be stronger when a firm exports to a more capital-abundant country.

Appendix A. Supplementary data

Supplementary data to this article can be found online at http://dx. doi.org/10.1016/j.jinteco.2013.11.003.

References

Ahn, J., Khandelwal, A., Wei, S.J., 2011. The role of intermediaries in facilitating trade. J. Int. Econ. 84, 73–85.

- Antweiler, W., Trefler, D., 2002. Increasing returns and all that: a view from trade. Am. Econ. Rev. 92, 93–119.
- Arkolakis, C., Muendler, M., 2011. The extensive margin of exporting products: a firmlevel analysis. Working Paper. Yale University.
- Aw, B., Chung, S., Roberts, M., 2000. Productivity and turnover in the export market: micro-level evidence from the Republic of Korea and Taiwan China. World Bank Econ. Rev. 14, 65–90.
- Aw, B., Roberts, M., Xu, D., 2011. R&D investment, exporting, and productivity dynamics. Am. Econ. Rev. 101, 1312–1344.
- Bastos, P., Silva, J., 2010. The quality of a firm's exports: where you export to matters. J. Int. Econ. 82, 99–111.
- Bernard, A., Jensen, B., 1997. Exporters, skill upgrading, and the wage gap. J. Int. Econ. 42, 3–31.
- Bernard, A., Jensen, B., 1999. Exceptional exporter performance: cause, effect, or both? J. Int. Econ. 47, 1–25.
- Bernard, A., Jensen, B., 2004. Why some firms export. Rev. Econ. Stat. 86, 561-568.
- Bernard, A., Wagner, J., 1997. Exports and success in German manufacturing. Weltwirtschaftliches Arch. 133, 134–157.
- Bernard, A., Eaton, J., Jensen, B., Kortum, S., 2003. Plants and productivity in international trade. Am. Econ. Rev. 93, 1268–1290.
- Bernard, A., Jensen, B., Schott, P., 2006. Survival of the best fit: exposure to low wage countries and the uneven growth of U.S. manufacturing plants. J. Int. Econ. 68, 219–237.
- Bernard, A., Jensen, B., Redding, S., Schott, P., 2007a. Firms in international trade. J. Econ. Perspect. 21, 105–130.
- Bernard, A., Redding, S., Schott, P., 2007b. Comparative advantage and heterogeneous firms. Rev. Econ. Stud. 74, 31–66.
- Bernard, A., Redding, S., Schott, P., 2010. Multiple-product firms and product switching. Am. Econ. Rev. 100, 70–97.
- Bernard, A., Redding, S., Schott, P., 2011. Multi-product firms and trade liberalization. Q. J. Econ. 126, 1271–1318.
- Brandt, L., Van Biesebroeck, J., Zhang, Y., 2012. Creative accounting or creative destruction? Firm-level productivity growth in Chinese manufacturing. J. Dev. Econ. 97, 339–351.
- Bustos, P., 2011. Trade liberalization, exports, and technology upgrading: evidence on the impact of MERCOSUR on Argentinian firms. Am. Econ. Rev. 101, 304–340.
- Clerides, S., Lach, S., Tybout, J., 1998. Is learning by exporting important? Micro-dynamic evidence from Colombia Mexico, and Morocco. Q. J. Econ. 113, 903–947.
- Dai, M., Maitra, M., Yu, M., 2011. Unexceptional exporter performance in China? The role of processing trade. Working paper. Peking University.
- De Loecker, J., 2007. Do exports generate higher productivity? Evidence from Slovenia. I. Int. Econ. 73, 69–98.
- Dhingra, S., 2013. Trading away wide brands for cheap brands. Am. Econ. Rev. (forthcoming).
- Eckel, C., Neary, J., 2010. Multi-product firms and flexible manufacturing in the global economy. Rev. Econ. Stud. 77, 188–217.
- Feenstra, R., Ma, H., 2008. Optimal choice of product scope for multiproduct firms under monopolistic competition. In: Helpman, E., Marin, D., Verdier, T. (Eds.), The Organization of Firms in a Global Economy. Harvard University Press.
- Harrigan, J., Reshef, A., 2012. Skill biased heterogeneous firms, trade liberalization and the skill premium. Working paper. University of Virginia.
- Harrigan, J., Ma, X., Shlychkov, V., 2012. Export prices of U.S. firms. Working paper. University of Virginia.
- Heckman, J., Ichimura, H., Todd, P., 1997. Matching as an econometric evaluation estimator. Rev. Econ. Stud. 65, 261–294.
- Hsieh, C., Klenow, P., 2009. Misallocation and manufacturing TFP in China and India. Q. J. Econ. 124, 1403–1448.

Jacovone, L. Javorcik, B., 2012. Getting ready: preparation for exporting. Working Paper. University of Oxford.

- International Study Group on Exports and Productivity, 2008. Exports and productivity - comparable evidence for 14 countries. Rev. World Econ. 144, 636-659
- Kasahara, H., Lapham, B., 2013. Productivity and the decision to import and export: theory and evidence. J. Int. Econ. 89, 297–316. Kee, H., Tang, H., 2012. Domestic value added in Chinese exports: firm-level evidence.
- Working paper. Johns Hopkins University.
- Kraay, A., 1999. Exports and economic performance: evidence from a panel of Chinese enterprises, Rev. Econ. Dev. 1-2, 183-207.
- Levinsohn, J., Petrin, A., 2003. Estimating production functions using inputs to control for unobservables. Rev. Econ. Stud. 70, 317-342.
- Lileeva, A., Trefler, D., 2010. Improved access to foreign markets raises plant-level productivity...for some plants. Q. J. Econ. 125, 1051-1099.
- Liu, R., 2010. Import competition and firm refocusing. Can. J. Econ. 43, 440-466.
- Lu, D., 2012. Exceptional exporter performance? Evidence from Chinese manufacturing firms. Working Paper. University of Rochester.
- Lucas, R., 1978. On the size distribution of business firms. Bell J. Econ. 9, 508-523.
- Manova, K., Yu, Z., 2012. Firms and credit constraints along the value-added chain: processing trade in China. NBER Working Paper, No.18561.
- Manova, K., Zhang, Z., 2012a. Export prices across firms and destinations. Q. J. Econ. 127, 379-436.

- Manova, K., Zhang, Z., 2012b, Multi-product firms and product quality, NBER Working Paper, No. 18637.
- Mayer, T., Melitz, M., Ottaviano, G., 2012. Market size, competition, and the product mix of exporters. Working paper. Harvard University. Melitz, M., 2003. The impact of trade on intra-industry reallocations and aggregate industry
- productivity. Econometrica 71, 1695–1725.
- Nocke, V., Yeaple, S., 2008. Globalization and the size distribution of multiproduct firms. NBER Working Paper, No.12322.
- Park, A., Yang, D., Shi, X., Jiang, Y., 2010. Exporting and firm performance: Chinese exporters and the Asian financial crisis. Rev. Econ. Stat. 92, 822–842.
- Qian, Z., Zhu, X., 2012. Misallocation or mismeasurement? Factor income shares and factor market distortions in China's manufacturing industries. Working paper. University of Toronto.
- Tang, H., Zhang, Y., 2012. Quality differentiation and trade intermediation. Working Paper.
- Johns Hopkins University. Van Biesebroeck, J., 2005. Exporting raises productivity in sub-Saharan African manufacturing firms. J. Int. Econ. 67, 373-391.
- Verhogeen, E., 2008. Trade, quality upgrading and wage inequality in the Mexican manufacturing sector. Q. J. Econ. 123, 489-530.
- World Bank, 1993. The East Asian Miracle: Economic Growth and Public Policy. Oxford University Press, New York.
- Yeaple, S., 2005. A simple model of firm heterogeneity, international trade, and wages. J. Int. Econ. 65, 1–20.